%================================================================
% LaTeX file with prefered layout for H1 paper drafts
% use: dvips -D600 file-name
%================================================================
\documentclass[12pt]{article}
\usepackage{epsfig}
\usepackage{graphicx}
\usepackage{amsmath}
\usepackage{hhline}
\usepackage{amssymb}
\usepackage{times}
\usepackage{cite}
\renewcommand{\topfraction}{1.0}
\renewcommand{\bottomfraction}{1.0}
\renewcommand{\textfraction}{0.0}
\newlength{\dinwidth}
\newlength{\dinmargin}
\setlength{\dinwidth}{21.0cm}
\textheight23.5cm \textwidth16.0cm
\setlength{\dinmargin}{\dinwidth}
\setlength{\unitlength}{1mm}
\addtolength{\dinmargin}{-\textwidth}
\setlength{\dinmargin}{0.5\dinmargin}
\oddsidemargin -1.0in
\addtolength{\oddsidemargin}{\dinmargin}
\setlength{\evensidemargin}{\oddsidemargin}
\setlength{\marginparwidth}{0.9\dinmargin}
\marginparsep 8pt \marginparpush 5pt
\topmargin -42pt
\headheight 12pt
\headsep 30pt \footskip 24pt
\parskip 3mm plus 2mm minus 2mm
%===============================title page=============================
\begin{document}
\begin{thebibliography}{99}


\bibitem{h1isol1998}
C.~Adloff {\it et al.} [H1 Collaboration],
Eur.\ Phys.\ J.\ C {\bf 5} (1998) 575 [hep-ex/9806009].

\bibitem{h1isol2003}
V.~Andreev {\it et al.} [H1 Collaboration],
Phys.\ Lett.\ B {\bf 561} (2003) 241 [hep-ex/0301030].

\bibitem{zeusisol2000}
J.~Breitweg {\it et al.} [ZEUS Collaboration],
Phys.\ Lett.\ B {\bf 471} (2000) 411 [hep-ex/9907023].

\bibitem{zeustop2003}
S.~Chekanov {\it et al.} [ZEUS Collaboration],
Phys.\ Lett.\ B {\bf 559} (2003) 153 [hep-ex/0302010].

\bibitem{zeusisolICHEP}
ZEUS Collaboration,
{\it ``Search for High-$P_{T}$ Isolated Leptons with ZEUS at HERA''},\\
contributed paper to {\it ICHEP06}, Moscow 2006.%, {\bf Abstract XX-XXXX}.

\bibitem{epvec}
U.~Baur, J.~A.~Vermaseren and D.~Zeppenfeld,
Nucl.\ Phys. B {\bf 375} (1992) 3.

\end{thebibliography}
  
% The rest
\newcommand{\pom}{{I\!\!P}}
\newcommand{\reg}{{I\!\!R}}
\newcommand{\slowpi}{\pi_{\mathit{slow}}}
%\newcommand{\gevsq}{\mathrm{GeV}^2}
\newcommand{\fiidiii}{F_2^{D(3)}}
\newcommand{\fiidiiiarg}{\fiidiii\,(\beta,\,Q^2,\,x)}
\newcommand{\n}{1.19\pm 0.06 (stat.) \pm0.07 (syst.)}
\newcommand{\nz}{1.30\pm 0.08 (stat.)^{+0.08}_{-0.14} (syst.)}
\newcommand{\fiidiiiful}{F_2^{D(4)}\,(\beta,\,Q^2,\,x,\,t)}
\newcommand{\fiipom}{\tilde F_2^D}
\newcommand{\ALPHA}{1.10\pm0.03 (stat.) \pm0.04 (syst.)}
\newcommand{\ALPHAZ}{1.15\pm0.04 (stat.)^{+0.04}_{-0.07} (syst.)}
\newcommand{\fiipomarg}{\fiipom\,(\beta,\,Q^2)}
\newcommand{\pomflux}{f_{\pom / p}}
\newcommand{\nxpom}{1.19\pm 0.06 (stat.) \pm0.07 (syst.)}
\newcommand {\gapprox}
   {\raisebox{-0.7ex}{$\stackrel {\textstyle>}{\sim}$}}
\newcommand {\lapprox}
   {\raisebox{-0.7ex}{$\stackrel {\textstyle<}{\sim}$}}
\def\gsim{\,\lower.25ex\hbox{$\scriptstyle\sim$}\kern-1.30ex%
\raise 0.55ex\hbox{$\scriptstyle >$}\,}
\def\lsim{\,\lower.25ex\hbox{$\scriptstyle\sim$}\kern-1.30ex%
\raise 0.55ex\hbox{$\scriptstyle <$}\,}
\newcommand{\pomfluxarg}{f_{\pom / p}\,(x_\pom)}
\newcommand{\dsf}{\mbox{$F_2^{D(3)}$}}
\newcommand{\dsfva}{\mbox{$F_2^{D(3)}(\beta,Q^2,x_{I\!\!P})$}}
\newcommand{\dsfvb}{\mbox{$F_2^{D(3)}(\beta,Q^2,x)$}}
\newcommand{\dsfpom}{$F_2^{I\!\!P}$}
\newcommand{\gap}{\stackrel{>}{\sim}}
\newcommand{\lap}{\stackrel{<}{\sim}}
\newcommand{\fem}{$F_2^{em}$}
\newcommand{\tsnmp}{$\tilde{\sigma}_{NC}(e^{\mp})$}
\newcommand{\tsnm}{$\tilde{\sigma}_{NC}(e^-)$}
\newcommand{\tsnp}{$\tilde{\sigma}_{NC}(e^+)$}
\newcommand{\st}{$\star$}
\newcommand{\sst}{$\star \star$}
\newcommand{\ssst}{$\star \star \star$}
\newcommand{\sssst}{$\star \star \star \star$}
\newcommand{\tw}{\theta_W}
\newcommand{\sw}{\sin{\theta_W}}
\newcommand{\cw}{\cos{\theta_W}}
\newcommand{\sww}{\sin^2{\theta_W}}
\newcommand{\cww}{\cos^2{\theta_W}}
\newcommand{\trm}{m_{\perp}}
\newcommand{\trp}{p_{\perp}}
\newcommand{\trmm}{m_{\perp}^2}
\newcommand{\trpp}{p_{\perp}^2}
\newcommand{\alp}{\alpha_s}

\newcommand{\alps}{\alpha_s}
\newcommand{\sqrts}{$\sqrt{s}$}
\newcommand{\LO}{$O(\alpha_s^0)$}
\newcommand{\Oa}{$O(\alpha_s)$}
\newcommand{\Oaa}{$O(\alpha_s^2)$}
\newcommand{\PT}{p_{\perp}}
\newcommand{\JPSI}{J/\psi}
\newcommand{\sh}{\hat{s}}
%\newcommand{\th}{\hat{t}}
\newcommand{\uh}{\hat{u}}
\newcommand{\MP}{m_{J/\psi}}
%\newcommand{\PO}{\mbox{l}\!\mbox{P}}
\newcommand{\PO}{I\!\!P}
\newcommand{\xbj}{x}
\newcommand{\xpom}{x_{\PO}}
\newcommand{\ttbs}{\char'134}
\newcommand{\xpomlo}{3\times10^{-4}}  
\newcommand{\xpomup}{0.05}  
\newcommand{\dgr}{^\circ}
\newcommand{\pbarnt}{\,\mbox{{\rm pb$^{-1}$}}}
\newcommand{\gev}{\,\mbox{GeV}}
\newcommand{\WBoson}{\mbox{$W$}}
\newcommand{\fbarn}{\,\mbox{{\rm fb}}}
\newcommand{\fbarnt}{\,\mbox{{\rm fb$^{-1}$}}}
%
% Some useful tex commands
%
\newcommand{\qsq}{\ensuremath{Q^2} }
\newcommand{\gevsq}{\ensuremath{\mathrm{GeV}^2} }
\newcommand{\et}{\ensuremath{E_t^*} }
\newcommand{\rap}{\ensuremath{\eta^*} }
\newcommand{\gp}{\ensuremath{\gamma^*}p }
\newcommand{\dsiget}{\ensuremath{{\rm d}\sigma_{ep}/{\rm d}E_t^*} }
\newcommand{\dsigrap}{\ensuremath{{\rm d}\sigma_{ep}/{\rm d}\eta^*} }
% Journal macro
\def\Journal#1#2#3#4{{#1} {\bf #2} (#3) #4}
\def\NCA{\em Nuovo Cimento}
\def\NIM{\em Nucl. Instrum. Methods}
\def\NIMA{{\em Nucl. Instrum. Methods} {\bf A}}
\def\NPB{{\em Nucl. Phys.}   {\bf B}}
\def\PLB{{\em Phys. Lett.}   {\bf B}}
\def\PRL{\em Phys. Rev. Lett.}
\def\PRD{{\em Phys. Rev.}    {\bf D}}
\def\ZPC{{\em Z. Phys.}      {\bf C}}
\def\EJC{{\em Eur. Phys. J.} {\bf C}}
\def\CPC{\em Comp. Phys. Commun.}


\begin{titlepage}

\begin{center}
%{\it {\large version 1.08 of \today}} \\[.3em]
\begin{small}
\begin{tabular}{llrr}
Submitted to & & &
\epsfig{file=/h1/www/images/H1logo_bw_small.epsi,width=2.cm} \\[.2em] \hline
%\epsfig{file=H1logo_bw_small.epsi,width=2.cm} \\[.2em] \hline
\multicolumn{4}{l}{{\bf
                33rd International Conference
                on High Energy Physics, ICHEP06},
                July~26,~2006,~Moscow} \\
                 & Abstract:        & {\bf }    &\\
                 & Parallel Session & {\bf Beyond the Standard Model}   &\\ \hline
 & \multicolumn{3}{r}{\footnotesize {\it www-h1.desy.de/h1/www/publications/conf/conf\_list.html}} \\[.2em]
\end{tabular}
\end{small}
\end{center}

% \noindent
% Date:               \\
% Version:            \\
% Editors:            \\
% Referees:           \\
% Comments by         
%\begin{flushright}
%H1prelim-04-063\\
%July 28, 2004
%\end{flushright}
\vspace{2cm}

\begin{center}
\begin{Large}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
{\bf \boldmath Search for Events with Isolated Leptons and Missing Transverse Momentum}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\vspace{2cm}

H1 Collaboration

\end{Large}
\end{center}

\vspace{1cm}

\begin{abstract}
\noindent

The search for events containing isolated leptons (electrons or muons) and
missing transverse momentum produced in $e^\pm p$ collisions is performed with
the H1 detector at HERA in the period 1994--2006. The analysed data sample corresponds
to an integrated luminosity of 341~pb$^{-1}$, which includes 223~pb$^{-1}$ of data
from the new HERA~II phase. A total of 46 events are observed in the
complete HERA data, compared to a Standard Model (SM) prediction of 43.0~$\pm$~6.0.
%
At large hadronic transverse momentum $P_{T}^{X} >$~25~GeV, a total of 18 events are
observed compared to a SM prediction of 11.5~$\pm$~1.8. In this region, 15 events are
observed in the $e^{+}p$ data compared to a SM prediction of 4.6~$\pm$~0.8, whereas in
the $e^{-}p$ data 3 events are observed compared to a SM prediction of 6.9~$\pm$~1.0.
%
A new comparison of the efficiency of the H1 and ZEUS experiments to detect events
containing isolated electrons or muons and missing transverse momentum is also presented.

\end{abstract}

\vspace{1.5cm}

\end{titlepage}

\newpage

\pagestyle{plain}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Introduction}
\label{sec:intro}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


Events containing a high $P_{T}$ isolated electron or muon associated
with missing transverse momentum have been observed at
HERA\cite{h1isol1998,h1isol2003,zeusisol2000,zeustop2003}. An excess of HERA~I
(1994--2000) data events compared to the SM prediction at large hadronic
transverse momentum $P_{T}^{X}$ was reported by the H1
Collaboration\cite{h1isol2003}. In the HERA~I data, 11 electron events and 8
muon events are observed compared to a SM expectation of 11.5~$\pm$~1.5 and
2.9~$\pm$~0.5 respectively. At large hadronic transverse momentum
$P_{T}^{X} >$~25~GeV, 5 electron events and 6 muon events are observed compared
to a SM expectation of 1.8~$\pm$~0.3 and 1.7~$\pm$~0.3 respectively.
The dominant SM contribution is from the production of real $W$ boasons.
The HERA~I data sample has an integrated luminosity of 104.7~pb$^{-1}$
in $e^{+}p$ scattering and 13.6~pb$^{-1}$ in $e^{-}p$ scattering. Presented
here is an update of the published H1 analysis, which includes
53~pb$^{-1}$ of $e^{+}p$ data and 170 pb$^{-1}$ of $e^{-}p$ data accumulated
at HERA~II from October 2003 to June 2006.




The observed H1 excess in the HERA~I data, most of which was taken in $e^{+}p$
collisions, was not confirmed by the ZEUS Collaboration, although it should be noted
that the anlaysis was performed in a slightly different phase space\cite{zeustop2003}.
The ZEUS analysis has been redone in a kinematic region more similar to the H1 selection
criteria \cite{zeusisolICHEP}. A comparison of the efficiency of the current
H1 and ZEUS analyses to detect events containing isolated electrons or muons
and missing transverse momentum is presented in section \ref{sec:effcomp}.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Standard Model Processes}
\label{sec:smproc}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


Signal events are defined as those that contain a prominent high $P_{T}$ isolated
lepton accompanied by genuine, large missing transverse momentum. The main SM
process that may produce events with this topology is the production of real $W$
bosons via photoproduction with subsequent leptonic decay:
$ep \rightarrow eW^{\pm}$($\rightarrow l\nu$)$X$. Additional, smaller contributions
arise from the production of $W$ bosons via the charged current process
$ep \rightarrow \nu$$W^{\pm}$($\rightarrow l\nu$)$X$ and the production of $Z^{0}$
bosons with subsequent decay to neutrinos
$ep \rightarrow eZ^{0}$($\rightarrow \nu\bar{\nu}$)$X$,
which only contributes to the electron channel.

The main SM background to events containing isolated electrons arises from neutral
current (NC) events, where fake missing transverse momentum is produced by a
fluctuation in the detector response, and from charged current (CC) events, where
a particle in the hadronic final state or a radiated photon is interpreted as
the isolated electron. For events containing an isolated muon the main SM
background arises from lepton pair (LP) events, where one muon escapes detection
and mismeasurement causes the apparent missing transverse momentum, and from CC
events where, similarly to the electron channel, a particle in the hadronic final
state is interpreted as the isolated muon.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Event Selection}
\label{sec:sel}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


The HERA~II event selection is identical to that used in the HERA~I
analysis and is detailed in table~1.%\ref{tab:cutstable}.
The main selection requirements are as follows: The lepton should have high
transverse momentum $P_{T}^{l}$, be observed in the central region of the
detector, defined by the polar angle $\theta_{l}$ and be isolated with
respect to jets and other tracks in the event. The isolation of the
lepton is quantified using the distances in $\eta-\phi$ space to the
nearest jet $D_{\rm jet}$ and nearest track $D_{\rm track}$. The event should
contain a large transverse momentum imbalance, both measured in the
calorimeter, $P_{T}^{\rm calo}$ and over the complete event,
$P_{T}^{\rm miss}$.


In order to reduce background contributions, further
cuts are then applied using variables sensitive to the presence of
high energy undetected particles in the event: The azimuthal balance
of the event $\frac{V_{ap}}{V_p}$, the difference in azimuthal angle
between the lepton and the hadronic system, $\Delta\phi_{l-X}$ and the
longitudinal momentum imbalance $\delta_{\rm miss}$. Additionally, muon events
are required to have a minimum $P_{T}^{X}$ and electron events
with low values of $P_{T}^{\rm calo}$ are required to have large $\zeta^{2}_{l}$,
which is equivalent to the four momentum transfer squared $Q^{2}$ in NC events.
To ensure that the two lepton channels are exclusive, electron events must contain
no isolated muons. Further details on these variables and the event selection
can be found in \cite{h1isol2003}.


\begin{table}[t]
\begin{center}
\begin{tabular}{|c||c|c|} \hline
 Variable  & Electron  & Muon \\ \hline
\hline
$\theta_{l}$ &  \multicolumn{2}{c|}{5$^\circ<\theta_l<$140$^\circ$}\\
\hline
$P_{T}^{l}$ & \multicolumn{2}{c|}{$>$ 10 GeV}\\
\hline
$P_{T}^{\rm calo}$ & \multicolumn{2}{c|}{$>$ 12 GeV}\\
\hline                                                   
$P_{T}^{\rm miss}$ & \multicolumn{2}{c|}{$>$ 12 GeV}\\
\hline
$P_{T}^{X}$ &  --  &$> 12$ GeV\\ 
\hline
$D_{\rm jet}$ & \multicolumn{2}{c|}{$>$ 1.0}\\
\hline
$D_{\rm track}$ & $>$ 0.5 for $\theta_e \ge$ 45$^\circ$ & $>$ 0.5 \\
\hline
$\zeta^{2}_{l}$ & $>$ 5000 GeV$^{2}$ for $P_{T}^{\rm calo} <$ 25 GeV & --\\
\hline
$\frac{V_{ap}}{V_p}$ & $<$ 0.5 ($<$ 0.15 for $P_{T}^{e} <$ 25 GeV) &
$<$ 0.5 ($<$ 0.15 for $P_{T}^{\rm calo} <$ 25 GeV)\\
\hline
$\Delta\phi_{l-X}$ & $<$ 160$^\circ$ & $<$ 170$^\circ$ \\
\hline
\# isolated $\mu$ & 0 & 1\\
\hline
$\delta_{\rm miss}$ & $>$ 5 GeV $^\dagger$ & -- \\
\hline
\end{tabular}
\begin{center}
  $^\dagger$ {\it if only one $e$ candidate is detected, which has the
    same charge as the beam lepton.}
\end{center}
\end{center}
\caption{Selection requirements for the electron and muon channels in the
	search for events with isolated leptons and missing transverse momentum.}
\label{tab:cutstable}
\end{table}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Results}
\label{sec:results}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


The analysis of the HERA II $e^{+}p$ data ($\cal{L}=$~53~pb$^{-1}$), which was
collected between October 2003 and August 2004, yields 9 electron events and
1 muon event compared to an expectation of 4.8~$\pm$~0.8 and 1.3~$\pm$~0.2
respectively. The majority of the SM prediction is due to signal processes,
dominated by real $W$ production. In the electron channel an excess of
events is observed in the data compared to the SM prediction at $P_T^X>25$~GeV,
where 5 events are observed compared to an expectation of 0.8~$\pm$~0.2. The
observation of 1 muon event in the data is in agreement with the SM event
yield. Figure \ref{fig:elasticevent} displays one of the HERA~II $e^{+}p$
electron events. It shows a high $P_{T}$ electron and large missing
transverse momentum. The absence of any reconstructed hadronic system
$X$ makes this event a candidate for elastic $W$ production. Figure
\ref{fig:highptxevent} shows a HERA~II $e^{+}p$ electron event with high
$P_{T}^{X}$. The same features are visible as in figure \ref{fig:elasticevent}, 
a high $P_{T}$ electron and large missing transverse momentum. However, also
present in this event is a prominent hadronic jet, giving rise to the large $P_T^X$.


The analysis of the HERA II $e^{-}p$ data ($\cal{L}=$~170~pb$^{-1}$), which was
collected between December 2004 and June 2006, yields 15 electron events and
2 muon events compared to an expectation of 17.7~$\pm$~2.5 and 4.7~$\pm$~0.7
respectively. Figure \ref{fig:highptxmuevent} shows a HERA~II $e^{-}p$ muon event
with high $P_{T}^{X}$. As for the HERA II $e^{+}p$ data the majority of the SM prediction
is due to signal processes, dominated by real $W$ production.
%However, as expected from the SM, the CC background is observed to be higher.
%The data excess observed at high $P_{T}^{X}$ in the HERA II $e^{+}p$ electron
%channel is not present in the $e^{-}p$ data, where 2 events are observed
%compared to a SM expectation of 3.5~$\pm$~0.6.


Table \ref{tab:resultstable} shows a summary of the analysis results for the
H1 $e^{+}p$ data ($\cal{L}=$~158~pb$^{-1}$), H1 $e^{-}p$ data
($\cal{L}=$~184~pb$^{-1}$) and the full HERA H1 $e^{\pm}p$ data set
($\cal{L}=$~341~pb$^{-1}$). In the H1 $e^{+}p$ data, 19 electron events and
9 muon events are observed compared to SM predictions of 14.6~$\pm$~2.0 and
3.9~$\pm$~0.6 respectively. Figure~\ref{fig:ptxhera_e+} shows the $P_{T}^{X}$
spectrum of the $e^{+}p$ data for the combined electron and muon channels.
In the H1 $e^{-}p$ data, 16 electron events and 2 muon events are observed
compared to SM predictions of 19.4~$\pm$~2.7 and 5.1~$\pm$~0.7 respectively.
The signal contribution is observed to be lower in the $e^{-}p$ data,
due to the higher SM background arising from CC events. Figure~\ref{fig:ptxhera_e-}
shows the $P_{T}^{X}$ spectrum of the $e^{-}p$ data for the combined electron
and muon channels.


In the complete H1 HERA data set, 35 electron events and 11 muon events are
observed compared to SM predictions of 34.0~$\pm$~4.7 and 9.0~$\pm$~1.4 respectively.
Figure~\ref{fig:ptxhera_e+e-} shows the $P_{T}^{X}$ spectrum of the $e^{\pm}p$ data
for the combined electron and muon channels. In 11 of the electron events the
lepton charge is measured as positive, in 8 events negative and is unmeasured in
the remaining 16 events. In 5 of the muon events the lepton charge is measured as
positive, in 5 events negative and is unmeasured in the remaining event. The main
signal process is expected to produce approximately two thirds positively charged
leptons. All background processes are expected to produce approximately similar
yields for both lepton charges except neutral current DIS, which produces
leptons with the same charge as the lepton beam.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{A Comparison of H1 and ZEUS Selection Efficiencies}
\label{sec:effcomp}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The isolated lepton selections employed by the H1 and ZEUS collaborations
are optimised for their respective detectors. The main difference in the 
selections is due to the cut on the lepton polar angle, which is
5~$^\circ < \theta_l <$~140~$^\circ$ for H1 and
17~$^\circ < \theta_e <$~86~$^\circ$,
17~$^\circ < \theta_\mu <$~115~$^\circ$ in the ZEUS analysis. Consequently,
the sensitivity of the detectors to the decay products from single $W$
production differs in some kinematic regions, which are defined in terms
of the polar angle $\theta_l$ and the transverse momentum of the final
state lepton $P_T^l$.

The efficiency of the two detectors is studied with a single $W$ production
Monte Carlo sample generated with EPVEC \cite{epvec}, using the respective
event selections from the H1 \cite{h1isol2003} and ZEUS \cite{zeusisolICHEP}
analyses. The efficiency is defined as the number of reconstructed
events passing the final selection divided by the number of generated events
in the kinematic region 5~$^\circ$~$<$~$\theta_l$~$<$~140~$^\circ$ and
$P_T^l >$~10~GeV.

Figure \ref{fig:h1-zeus}a shows the H1 and ZEUS efficiencies for detecting
electrons from single $W$ decay as a function of $P_{T}^{X}$. The
observed drop in the ZEUS efficiency in the lowest bin is due to the analysis
cut $P_{T}^{X} >$~12~GeV. At large $P_{T}^{X}$ the efficiencies are seen to
be flat and more comparable. The arrows indicate the position of the
observed data events. Figure \ref{fig:h1-zeus}b shows the efficiency as a
function of $\theta_{e}$ at large hadronic transverse momentum
$P_{T}^{X} >$~25~GeV, the kinematic region where an excess of data events
is observed by H1 over the SM prediction. Here the cut $P_{T}^{X} >$~25~GeV
is also made at the generator level. The $\theta_{e}$ distribution from EPVEC
is also shown. It can be seen that in the common polar angle range the
efficiencies of the H1 and ZEUS selection at large $P_{T}^{X}$ are similar.
Additionally, the majority of the H1 events observed at high $P_{T}^{X}$
lie within this common region. The more limited angular range in the ZEUS
analysis visible in figure \ref{fig:h1-zeus}b is the main reason for the
lower ZEUS efficiency visible at high $P_{T}^{X}$ in figure~\ref{fig:h1-zeus}a.

The efficiencies in the muon channel are similarly displayed in figures
\ref{fig:h1-zeus}c and \ref{fig:h1-zeus}d. As observed in the electron channel,
the efficiency as a function of $P_{T}^{X}$ is found to be comparable between
the H1 and ZEUS analyses and reasonably flat. It can also be seen that all
six high $P_{T}^{X}$ muon data events observed by H1 fall within the geometrical
acceptance of the ZEUS selection.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Summary}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The search for events containing isolated electrons or muons and large missing
transverse momentum produced in $e^{\pm}p$ collisions is performed with
the H1 detector at HERA in the period 1994--2006. The analysed data sample
corresponds to an integrated luminosity of 341 pb$^{-1}$. A total of 46 events
are observed in the data, compared to a SM prediction of 43.0~$\pm$~6.0.

At large hadronic transverse momentum $P_{T}^{X} >$ 25 GeV, a total of
18 events are observed compared to a SM prediction of 11.5~$\pm$~1.8.
In 9 events the lepton charge is positive, in 4 events negative and unmeasured
in the remaining 5 events. The probability for the SM expectation to fluctuate to
the observed number of events or more in the high $P_{T}^{X}$ domain for all data
is 0.0673 compared to 0.0015 for the HERA I data (the majority of which is
$e^{+}p$ data). For the $e^{+}p$ data alone, this probability is 0.0003.

The selection efficiencies of the H1 and ZEUS isolated lepton analyses have been
examined and are found to be compatible in the kinematic region where the two
analyses are directly comparable. It is also found that the majority
of the data events observed by H1 at $P_{T}^{X} >$ 25 GeV fall
into the region of common phase space.

The continued increase in luminosity expected from the HERA II programme in the
coming years will improve the precision of the measurement of events with
isolated leptons and missing transverse energy and will hopefully clarify the
excess observed by H1 in the $e^{+}p$ data at large hadronic transverse momentum.


\newpage

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newpage

\begin{figure}[p] 
  \begin{center}
      \includegraphics[width=16cm]{H1prelim-06-162.fig1.eps}
  \end{center}
\begin{picture} (0.,0.)
\setlength{\unitlength}{1.0cm}
\put (0.2,9.05){\bf\Large Event with $e+P_T^{miss}$ in HERA II $e^{+}p$ data }
\put (0.2,8.05){\bf\large $\mathbf P_T^e=47$~GeV, $\mathbf P_T^{miss}=47$~GeV}
\put (10.22,2.75){\bf\Large $e^+$} 
\end{picture}
\caption{Display of an event with an isolated electron and missing transverse
	momentum in the HERA II $e^{+}p$ data: a candidate for
	elastic $W$ production.}
\label{fig:elasticevent}
\end{figure} 
%
%
%
%
\begin{figure}[p] 
  \begin{center}
      \includegraphics[width=16cm]{H1prelim-06-162.fig2.eps}
  \end{center}
\begin{picture} (0.,0.)
\setlength{\unitlength}{1.0cm}
\put (0.2,9.05){\bf\Large Event with $e+P_T^{miss}$ in HERA II $e^{+}p$ data }
\put (0.2,8.05){\bf\large $\mathbf P_T^e=37$~GeV, $\mathbf P_T^{miss}=44$~GeV,
	$\mathbf P_T^{X}=29$~GeV}
\put (10.66,2.45){\bf\Large $e^+$} 
\put (14.04,3.40){\bf\Large $X$} 
\end{picture}
\caption{Display of an event with an isolated electron, missing transverse
	momentum and a prominent hadronic jet in the HERA II $e^{+}p$ data.}
\label{fig:highptxevent}
\end{figure} 

\clearpage

\begin{figure}[t] 
  \begin{center}
      \includegraphics[width=15cm]{H1prelim-06-162.fig3.eps}
  \end{center}
\begin{picture} (0.,0.)
\setlength{\unitlength}{1.0cm}
\put (0.2,9.05){\bf\Large Event with $\mu+P_T^{miss}$ in HERA II $e^{-}p$ data }
\put (0.2,8.05){\bf\large $\mathbf P_T^\mu=38$~GeV, $\mathbf P_T^{miss}=51$~GeV,
	$\mathbf P_T^{X}=24.7$~GeV}
\put (13.3,4.4){\bf\Large $X$}
\put (12.0,6.925){\bf\Large $\mu^{-}$}
\end{picture}
\caption{Display of an event with an isolated muon, missing transverse
	momentum and a prominent hadronic jet in the HERA II $e^{-}p$ data.}
\label{fig:highptxmuevent}
\end{figure} 


\begin{table}
 \renewcommand{\arraystretch}{1.6}
\begin{center}
  \begin{tabular}{|c|c|c|c|c|}
    \hline
    \multicolumn{2}{|c|}{} &
    Electron &
    Muon &
    Combined \\
    \multicolumn{2}{|c|}{\large H1 Preliminary} &
    obs./exp. &
    obs./exp. &
    obs./exp. \\
    \multicolumn{2}{|c|}{} &
    {\footnotesize (Signal contribution)} &
    {\footnotesize (Signal contribution)} &
    {\footnotesize (Signal contribution)} \\
    \hline
    \hline
    {\footnotesize 1994-2004 $e^{+} p$} &
    {\footnotesize Full Sample} &
    {\footnotesize 19 / 14.6 $\pm$ 2.0 (70\%)}&
    {\footnotesize  9 /  3.9 $\pm$ 0.6 (84\%)}&
    {\footnotesize 28 / 18.5 $\pm$ 2.6 (73\%)}\\
    \cline{2-5}
    {\footnotesize 158 pb$^{-1}$} &
    {\footnotesize $P_{T}^{X}~>25$~GeV} &
    {\footnotesize  9 /  2.3 $\pm$ 0.4 (80\%)}&
    {\footnotesize  6 /  2.3 $\pm$ 0.4 (84\%)}& 
    {\footnotesize 15 /  4.6 $\pm$ 0.8 (82\%)}\\
    \hline
    \hline
    {\footnotesize 1998-2006 $e^{-} p$} &
    {\footnotesize Full Sample} &
    {\footnotesize 16 / 19.4 $\pm$ 2.7 (65\%)}&
    {\footnotesize  2 /  5.1 $\pm$ 0.7 (78\%)}&
    {\footnotesize 18 / 24.4 $\pm$ 3.4 (68\%)}\\
    \cline{2-5}
    {\footnotesize 184 pb$^{-1}$} &
    {\footnotesize $P_{T}^{X}~>25$~GeV} &
    {\footnotesize  3 /  3.8 $\pm$ 0.6 (61\%)}&
    {\footnotesize  0 /  3.1 $\pm$ 0.5 (74\%)}& 
    {\footnotesize  3 /  6.9 $\pm$ 1.0 (67\%)}\\
    \hline
    \hline
    {\footnotesize 1994-2006 $e^{\pm} p$} &
    {\footnotesize Full Sample} &
    {\footnotesize 35 / 34.0 $\pm$ 4.7 (68\%)}&
    {\footnotesize 11 /  9.0 $\pm$ 1.4 (80\%)}&
    {\footnotesize 46 / 43.0 $\pm$ 6.0 (70\%)}\\
    \cline{2-5}
    {\footnotesize 341 pb$^{-1}$} &
    {\footnotesize $P_{T}^{X}~>25$~GeV} &
    {\footnotesize 12 /  6.1 $\pm$ 1.1 (66\%)}&
    {\footnotesize  6 /  5.4 $\pm$ 0.9 (77\%)}& 
    {\footnotesize 18 / 11.5 $\pm$ 1.8 (71\%)}\\
    \hline
  \end{tabular}
  \caption{Summary of the H1 results of searches for events with isolated electrons
	or muons and missing transverse momentum for the $e^{+}p$ data
	(${\cal L}=$~158~pb$^{-1}$), $e^{-}p$ data (${\cal L}=$~184~pb$^{-1}$) and
	the full HERA data set (${\cal L}=$~341~pb$^{-1}$). The results are shown
	for the full selected sample and for the subsample at large
	$P_{T}^{X}>25$~GeV. The number of observed events is compared to the SM
	prediction. The signal component of the SM expectation, dominated by real
	$W$ production, is given as a percentage in parentheses. The quoted errors
	contain statistical and systematic uncertainties added in quadrature.}
\label{tab:resultstable}
\end{center}
\end{table}


\begin{figure}[ht] 
  \begin{center}
      \includegraphics[width=15cm]{H1prelim-06-162.fig4.eps}
  \end{center}
  \caption{The hadronic transverse momentum distribution in the electron
	and muon channels combined: data ($e^{+}p$, ${\cal L}=$~158~pb$^{-1}$)
	is compared to the SM expectation (open histogram).  The signal
	component of the SM expectation, dominated by real $W$ production,
	is given by the hatched histogram. $\rm N_{Data}$ is the total number
	of data events observed, $\rm N_{SM}$ is the total SM expectation. The
    total error on the SM expectation is given by the shaded band.}
  \label{fig:ptxhera_e+}
\end{figure} 


\begin{figure}[ht] 
  \begin{center}
      \includegraphics[width=15cm]{H1prelim-06-162.fig5.eps}
  \end{center}
  \caption{The hadronic transverse momentum distribution in the electron and
	muon channels combined: data ($e^{-}p$, ${\cal L}=$~184~pb$^{-1}$)
	is compared to the SM expectation (open histogram).  The signal
	component of the SM expectation, dominated by real $W$ production,
	is given by the hatched histogram. $\rm N_{Data}$ is the total number
	of data events observed, $\rm N_{SM}$ is the total SM expectation. The
    total error on the SM expectation is given by the shaded band.}
  \label{fig:ptxhera_e-}
\end{figure} 


\begin{figure}[ht] 
  \begin{center}
      \includegraphics[width=15cm]{H1prelim-06-162.fig6.eps}
  \end{center}
  \caption{The hadronic transverse momentum distribution in the electron and
	muon channels combined: data ($e^{\pm}p$, ${\cal L}=$~341~pb$^{-1}$)
	is compared to the SM expectation (open histogram).  The signal
	component of the SM expectation, dominated by real $W$ production,
	is given by the hatched histogram. $\rm N_{Data}$ is the total number
	of data events observed, $\rm N_{SM}$ is the total SM expectation. The
    total error on the SM expectation is given by the shaded band.}
  \label{fig:ptxhera_e+e-}
\end{figure} 

\begin{figure}[ht]
  \begin{center}
    \includegraphics[width=.49\textwidth]{H1prelim-06-162.fig7a.eps}
    \includegraphics[width=.49\textwidth]{H1prelim-06-162.fig7b.eps}
    \includegraphics[width=.49\textwidth]{H1prelim-06-162.fig7c.eps}
    \includegraphics[width=.49\textwidth]{H1prelim-06-162.fig7d.eps}
  \end{center}
  \begin{picture} (0.,0.)
  \setlength{\unitlength}{1.0cm}
    \put (6.0,9.4){\bf\Large a)}
    \put (14.0,9.4){\bf\Large b)}
    \put (6.0,2.3){\bf\Large c)}
    \put (14.0,2.3){\bf\Large d)}
  \end{picture}
\caption{Comparisons of the efficiencies of the H1 and ZEUS analyses to select
	events containing isolated electrons or muons and missing transverse
	momentum. Figure a) displays the efficiency, defined as
	$N_{rec}$~/~$N_{gen}$, as a function of hadronic transverse momentum
	$P_{T}^{X}$ in the electron channel. Figure b) displays the efficiency in
	the electron channel at large $P_{T}^{X} >$~25~GeV as a function of
	electron polar angle. Figure c) displays the efficiency as a function of
	hadronic transverse momentum $P_{T}^{X}$ in the muon channel. Figure d)
	displays the efficiency in the muon channel at large $P_{T}^{X} >$~25~GeV
	as a function of muon polar angle. In all figures the H1 (ZEUS) efficiency
	is given by the solid (dashed) line. The data events observed by each
	experiment are indicated by the arrows. The $\theta_{l}$ distribution
	from EPVEC is also shown in figures b) and d).}
\label{fig:h1-zeus}
\end{figure}


\end{document}
